1. **Suggestion:** This is an interesting case report about the hypoxemia after radiofrequency ablation for atrial fibrillation complicated with tetralogy of Fallot. This manuscript is nicely structured and well written. I have one minor comment about this manuscript. Please consider the following comment. (Comment) 1. Page 15, Figure 5 Correct “occlader” to “occluder”.

**Reply:** We have revised the article as suggested.

Comment ID: 06116605

1. **Comment:** Title: Lacks clarity & comprehensiveness.

Hypoxemia is a known fact in Cyanotic heart disease, hence just mentioning hypoxemia is not meaningful.

**Suggestion:** Severe hypoxemia complication after radiofrequency ablation for atrial fibrillation complicated with in palliatively repaired Tetralogy of Fallot: a case report

**Reply:** We have revised the article as suggested.

2. **Comment:** Abstract: “The development of major cardiac arrhythmias increases over time, and most commonly includes atrial flutter or fibrillation (present in ≤ 20% of patients) or sustained ventricular tachycardia (present in ≤ 14% of patients)”

**Suggestion:** Patients with tetralogy of Fallot often have arrhythmias, common being one of which is atrial fibrillation.

**Reply:** We have revised the article as suggested.

3. **Comment:** Conclusion: Well known: “The presence of arrhythmias usually reflects hemodynamic deterioration from the right and/or left heart, and should be treated accordingly”.

**Suggestion:** This case alerts us to the need for proper complete further evaluation of
patients, including hemodynamics, with tetralogy of Fallot before radiofrequency ablation. This procedure may cause iatrogenic atrial septal injury, and one needs to be prepared to treat complications, to ensure success & patient safety.

Reply: Due to the word limit when submitting articles, we have condensed our conclusions.

4. Comment: Core Tip: Patients with clinically significant arrhythmias may benefit from catheter ablation. Suggestion: This case has important reference significance for the selection of treatment decision proper preparation and perioperative management of atrial fibrillation under special circumstances.

Reply: We have revised the article as suggested.

Question: Is there any other option preferred over catheter ablation?
Reply: Multiple studies have shown that atrial fibrillation ablation is safe in congenital heart disease, and atrial fibrillation can be controlled in most patients regardless of complexity.(1-4) However, for more therapeutic measures, such as drugs and pulsed field ablation, relevant randomized controlled trials are still lacking and need to be further explored.

Reply: What we want to express is that the patient is in a state of mental distress, mainly manifested by lethargy, lethargy and so on.

Reference
